Lifestyle changes during the coronavirus disease 2019 pandemic impact metabolic dysfunction–associated fatty liver disease

メタデータ	言語: English
	出版者: Wiley
	公開日: 2023-03-31
	キーワード (Ja): ボディマス指数,
	非アルコール性脂肪肝疾患
	キーワード (En): body mass index, non-alcoholic fatty
	liver disease, obesity, physical activity
	作成者: 藤井, 英樹, 中村, 直俊, 福本, 真也, 木村, 達郎,
	中野, 朱美, 灘谷, 祐二, 田内, 幸枝, 西居, 由布子, 高嶋,
	信吾, 鎌田, 佳宏, 渡邉, 俊雄, 河田, 則文
	メールアドレス:
	所属: Osaka City University, Nagoya University, Osaka
	City University, Osaka City University, Osaka City
	University, Osaka City University, Osaka City University,
	Osaka City University, Osaka City University, Osaka
	University, Osaka City University, Osaka City University
URL	https://ocu-omu.repo.nii.ac.jp/records/2020090

Lifestyle changes during the coronavirus disease 2019 pandemic impact metabolic dysfunction—associated fatty liver disease

Hideki Fujii, Naotoshi Nakamura, Shinya Fukumoto, Tatsuo Kimura, Akemi Nakano, Yuji Nadatani, Yukie Tauchi, Yuuko Nishii, Shingo Takashima, Yoshihiro Kamada, Toshio Watanabe, Norifumi Kawada

Citation	Liver International. 42(5); 995-1004.			
Issue Date	2022-05			
Version of Record	2022-03-18			
Туре	Journal Article			
Textversion	Author			
	◇新型コロナウイルス感染症(COVID-19)のパンデミック前後で健康診断受診者の			
TT: alt li alt 4a	データを解析した結果、代謝異常関連脂肪肝(MAFLD)の患者数が増加しているこ			
Hignlights	とが判明。			
	◇MAFLD 患者増加には夜食・飲酒・欠食等が関連。			
	This is the peer reviewed version of the following article: Liver International,			
	Vol.42, Issu.5, 995-1104. which has been published in final form at			
	https://doi.org/10.1111/liv.15158. This article may be used for non-commercial			
	purposes in accordance with Wiley Terms and Conditions for Use of Self-Archived			
	Versions. This article may not be enhanced, enriched or otherwise transformed			
Rights	into a derivative work, without express permission from Wiley or by statutory			
	rights under applicable legislation. Copyright notices must not be removed,			
	obscured or modified. The article must be linked to Wiley's version of record on			
	Wiley Online Library and any embedding, framing or otherwise making available			
	the article or pages thereof by third parties from platforms, services and websites			
	other than Wiley Online Library must be prohibited.			
Supporting	Supporting Information is available at <u>https://doi.org/10.1111/liv.15158</u>			
Information				
DOI	10.1111/liv.15158			

Self-Archiving by Author(s)

Placed on: Osaka City University Repository

Fujii, H., Nakamura, N., Fukumoto, S., Kimura, T., Nakano, A., Nadatani, Y., Tauchi, Y., Nishii, Y., Takashima, S., Kamada, Y., Watanabe, T., & Kawada, N. (2022). Lifestyle changes during the coronavirus disease 2019 pandemic impact metabolic dysfunction-associated fatty liver disease. *Liver International.* 42, 995-1004. https://doi.org/10.1111/liv.15158

	研究グループは、健診受診者の COVID-19 パンデミック前後における臨床検査データや
	生活習慣に関するアンケート結果を解析し、COVID-19 パンデミック後に MAFLD 患者が
	増加したこと、また、MAFLD 患者増加には夜食・飲酒・欠食(1 日 2 食のみ)が関与する
概要	ことを明らかにしました。
	・新型コロナウイルス感染症のパンデミックで脂肪肝が増加ー背景に夜食や欠食などの生活習慣
	の変化'. 大阪市立大学. <u>https://www.osaka-cu.ac.jp/ja/news/2021/220324</u> (参照 2022/0/03/24)

Article type : Original Article Editor : Luca Valenti

Lifestyle changes during the coronavirus disease 2019 pandemic impact metabolic dysfunction–associated fatty liver disease Short title: COVID-19 lifestyle changes affect MAFLD

Hideki Fujii^{1*}, Naotoshi Nakamura², Shinya Fukumoto¹, Tatsuo Kimura¹, Akemi Nakano¹, Yuji Nadatani¹, Yukie Tauchi¹, Yuuko Nishii¹, Shingo Takashima¹, Yoshihiro Kamada³, Toshio Watanabe¹, and Norifumi Kawada⁴

¹Department of Premier Preventive Medicine, Graduate School of Medicine, Osaka City University, Osaka, Japan

²Center for Mathematical Modeling and Data Science, Osaka University, Osaka, Japan

³Department of Advanced Metabolic Hepatology, Osaka University, Graduate School of

Medicine, Suita, Osaka, Japan

⁴Department of Hepatology, Graduate School of Medicine, Osaka City University, Osaka, Japan

*Corresponding author: Hideki Fujii, M.D. Ph.D. Department of Premier Preventive Medicine, Graduate School of Medicine, Osaka City University, 1-4-3, Asahimachi, Abeno, Osaka 545–8585, Japan Tel: +81-6-6645-3905

Email: rolahideki@med.osaka-cu.ac.jp

This article has been accepted for publication and undergone full peer review but has not been through the copyediting, typesetting, pagination and proofreading process, which may lead to differences between this version and the <u>Version of Record</u>. Please cite this article as <u>doi:</u> <u>10.1111/liv.15158</u>

Word counts for main body (from title page to the beginning of the references): 3742 Total number of figure and tables: 6

List of abbreviations:

ALT, alanine aminotransferase; AST, aspartate aminotransferase; BMI, body mass index; BP, blood pressure; CI, confidence interval; COVID-19, coronavirus disease 2019; DM, diabetes mellitus; FIB-4, Fibrosis-4; GGT, gamma-glutamyl transferase, HbA1c, glycosylated hemoglobin A1c; HBsAg, hepatitis B surface antigen; HCV-Ab, anti-hepatitis C virus antibody; HDL-C, high-density lipoprotein-cholesterol; HR, hazard ratio; MAFLD, metabolic dysfunction–associated fatty liver disease; NAFLD, nonalcoholic fatty liver disease; TG, triglycerides; WC, waist circumference.

Disclosures/Conflicts of interest: All authors declare that they have no conflicts of interest to disclose.

Grant support:

This work was supported by a Grant-in-Aid for Scientific Research (C) from JSPS KAKENHI 17K09437 (to HF).

Writing assistance: Not applicable.

Author Contributions: Writing: HF; data collection: SF, TK, AN, YN, YN, ST, and TW; statistical analyses: HF and NN; revision and editing: HF, YK, and NK; acceptance of the final version: all authors.

Data transparency statement: Data are available on reasonable request to the corresponding author.

Acknowledgments:

The authors are grateful to Tetsuya Okawa for technical assistance. The authors also thank BioScienceWriters (https://www.biosciencewriters.com/) for English language editing.

Abstract

BACKGROUND & AIMS: The coronavirus disease 2019 (COVID-19) pandemic precipitated lifestyle changes. We aimed to clarify whether COVID-19–induced lifestyle changes affected the development of metabolic dysfunction–associated fatty liver disease (MAFLD).

METHODS: This retrospective longitudinal study included 973 participants who underwent health checkups between 2018 and 2020. We used data from the MedCity21 health examination registry. Participants' clinical characteristics and lifestyle habits were investigated. Independent lifestyle predictors of MAFLD development before the pandemic (2018–2019) and during the pandemic (2019–2020) were identified using logistic regression analysis.

RESULTS: In 2018, 261 (27%) patients were diagnosed with MAFLD. Before the pandemic, 22 patients developed new MAFLD. During this time, routine late-night meals were identified as an independent lifestyle predictor of MAFLD development (hazard ratio [HR] 2.54, 95% confidence interval [CI] 1.02–6.36, P=0.046). In contrast, 44 patients developed new MAFLD during the pandemic. During this time, higher daily alcohol intake was identified as an independent lifestyle predictor of MAFLD development (HR 1.03, 95% CI 1.01–1.05, P=0.008). In participants aged <60 years, daily alcohol intake and the proportion of participants who ate 2 times/day were significantly higher in patients who developed MAFLD during the pandemic than in those who did not. In participants aged \geq 60 years, no lifestyle habits were associated with MAFLD development before or during the pandemic.

CONCLUSIONS: New MAFLD diagnoses increased during the COVID-19 pandemic. Changes in lifestyle factors, particularly in those aged <60 years, must be monitored and addressed as the pandemic continues.

Keywords: physical activity, body mass index, obesity, nonalcoholic fatty liver disease

Lay summary:

The impact of the lifestyle changes associated with the COVID-19 pandemic on metabolic dysfunction-associated fatty liver disease (MAFLD) remains unknown. Late-night meals before the coronavirus disease 2019 (COVID-19) pandemic and daily alcohol intake during the COVID-19 pandemic were extracted as independent lifestyle predictors of MAFLD development. These results are relevant for patient lifestyle counseling to prevent the increasing number of patients with MAFLD in the COVID-19 pandemic.

Introduction

The coronavirus disease 2019 (COVID-19) outbreak in December 2019 spread rapidly throughout the world and was declared a global pandemic by the World Health Organization in March 2020.¹ The Japanese government declared a state of emergency on April 7, 2020 because of a rapid increase in the number of patients affected by COVID-19. Under this declaration, nonessential businesses, schools, and recreational facilities were urged to close, while individuals were strongly encouraged to stay at home except for essential activities. The COVID-19 crisis forced lifestyle changes, the so-called "new normal." For example, many people now work from home and have little contact with people other than their family members. These changes potentially led to less physical activity, altered daily life rhythms, and unhealthy lifestyles.² They also may have resulted in increases in body weight and body mass index (BMI), potentially increasing the incidence of obesity and obesity-related disorders.^{3, 4}

Because of widespread obesity, nonalcoholic fatty liver disease (NAFLD) is one of the leading causes of liver disease worldwide. The prevalence of NAFLD globally is currently estimated to be approximately 25%.⁵ NAFLD comprises a broad spectrum of diseases ranging from nonalcoholic fatty liver to nonalcoholic steatohepatitis to cirrhosis and/or hepatocellular

carcinoma.^{6, 7} In 2020, an international consensus panel suggested that NAFLD should be renamed metabolic dysfunction–associated fatty liver disease (MAFLD) and proposed new information regarding its diagnosis.^{8, 9}

The impact of "new normal" lifestyles on the incidence of MAFLD is unknown. We hypothesized that new lifestyles resulting from the COVID-19 pandemic affect the development of MAFLD. Thus, we aimed to determine whether lifestyle changes associated with the COVID-19 pandemic increased the incidence of MAFLD and to identify specific lifestyle changes associated with the development of MAFLD.

Methods

Study design and population

This retrospective longitudinal study of MAFLD was based on an analysis of data obtained from the MedCity21 health examination registry, and was performed in Osaka, the second-largest city in Japan. The MedCity21 health examination registry protocol included regular comprehensive health assessments and was conducted in full accordance with the tenets of the Declaration of Helsinki (approval no. 2927).¹⁰ The protocol for the current study was approved by the ethics committee of the Graduate School of Medicine, Osaka City University (approval no. 2021-027; April 26, 2021). Written informed consent was waived because of the study's retrospective nature. To allow participants the opportunity to refuse to participate, we provided a web-based opt-out option.

We selected all individuals who underwent abdominal ultrasonography in 2018 and analyzed those who returned for routine follow-up until 2020. Repeat visitors who underwent medical checkups during period 1 (July 1–December 31, 2018), period 2 (July 1–December 31, 2019), and period 3 (July 1–December 31, 2020) were included (Figure 1). We defined the

duration from period 1 to period 2 as the time before the COVID-19 pandemic and the duration from period 2 to period 3 as the time during the COVID-19 pandemic. The exclusion criteria were as follows: lack of data regarding hepatitis B surface antigen (HBsAg)/anti-hepatitis C antibody (HCV-Ab), serologic positivity for HBsAg and HCV-Ab, or any missing data components. Health checkup was postponed for examinees who were PCR positive for COVID-19 within 4 weeks from the date of examination or had symptoms (cough, fever, etc.) suspicious for COVID-19 infection on the date of examination. While analyzing lifestyle associations with the development of MAFLD, we also performed subgroup analyses according to age because lifestyle patterns differ between younger and older populations.^{11, 12}

Clinical assessment

Data regarding the participants' clinical and demographic characteristics (age, sex, height, body weight, and blood pressure [BP]) and laboratory parameters were measured and recorded during all medical checkups. BMI was calculated as weight (kg) / height (m)². Waist circumference (WC) was measured with a non-stretchable tape at the level of the umbilicus with the participant in the standing position during the late phase of expiration. All data related to lifestyle habits (smoking history, physical exercise frequency, sleeping duration, meals per day, and late-night meals) and medical history were obtained using a questionnaire. A routine late-night meal was defined as having dinner within 2 hours before bedtime at least 3 times per week. The presence or absence of diabetes mellitus (DM), hypertension, and dyslipidemia was recorded; these diseases were collected and analyzed using standard laboratory procedures to obtain values for the complete blood count, aspartate aminotransferase (AST), alanine aminotransferase (ALT), albumin, gamma-glutamyl transferase (GGT), total cholesterol, low-density lipoprotein-cholesterol,

high-density lipoprotein-cholesterol (HDL-C), triglycerides (TG), fasting plasma glucose, and glycosylated hemoglobin A1c (HbA1c). Lumipulse HBsAg and HCV assays (Fujirebio Inc., Tokyo, Japan) were used to assess serologic markers, including HBsAg and anti-HCV-Ab. The Fibrosis-4 (FIB-4) index was calculated using available parameters.¹⁶

Fatty liver diagnosis

Fatty liver was diagnosed via abdominal ultrasonography using a Toshiba Aplio 500 Ultrasound Machine (Toshiba Medical Systems Corporation, Ohtawara, Japan). Abdominal ultrasonography was performed by experienced sonographers who were registered with the Japan Society of Ultrasonics in Medicine. Hepatic steatosis was semi-quantified according to the criteria described by Hamaguchi et al., which are based on the presence of hepatorenal echo contrast, a bright liver, deep attenuation, and vessel blurring.¹⁷

Alcohol intake screening and definition of alcoholic liver disease

Daily alcohol consumption was calculated in grams using our modified template.^{18, 19} Frequency of alcohol intake was classified as 1 day per week, 3 days per week, or 7 days per week. Each participant's average alcohol consumption was classified as 10, 30, 50, or 70 g per day on days when alcohol was consumed. Daily alcohol consumption (g/day) was calculated as follows: (frequency of alcohol intake × average alcohol consumption [g]) / 7. Habitual alcohol intake was defined as alcohol intake of 1–59 g/day.

Diagnostic criteria and definition of MAFLD and NAFLD

MAFLD was defined as hepatic steatosis identified on abdominal ultrasonography and the presence of any 1 of these 3 conditions: overweight/obesity, DM, or evidence of metabolic

dysregulation. Metabolic dysregulation was defined as the presence of 2 or more of the following: WC \geq 90 cm for men and \geq 80 cm for women; BP \geq 130/85 mmHg or current specific antihypertensive drug treatment; TG \geq 150 mg/dL or current specific hypolipidemic drug treatment; HDL-C < 40 mg/dL for men and <50 mg/L for women; or prediabetes (fasting glucose 100–125 mmol/L or HbA1c 5.7%–6.4%).⁸ We did not use the homeostasis model assessment of the insulin resistance score or high-sensitivity C-reactive protein level during this study. NAFLD was defined as ultrasonographic evidence of fatty liver disease and lack of excessive alcohol consumption (\geq 30 g/day for men and > 20 g/day for women).^{6, 7} We defined "MAFLD development" as no MAFLD in the first year but the presence of MALD in the following year. "MAFLD resolution" was defined as the presence of MAFLD in the first year but no MALD in the following year.

Statistical analyses

Continuous variables were expressed as median (range), mean (standard deviation [SD]), or number (%). Categorical variables were expressed as frequency and percentage. Differences between groups were analyzed using the Wilcoxon rank-sum test for continuous variables and the Kruskal–Wallis test for categorical variables. Multiple comparisons were performed using Steel's multiple comparison test. A logistic regression model was used to identify independent predictors associated with MAFLD development.

Univariate (unadjusted) and multivariate (adjusted) hazard ratio (HR) estimates of the risk of MAFLD development were calculated to control for the effects of potential risk factors (confounders), including clinically important variables, such as age, sex, daily alcohol intake, BMI, Δ BMI, and ALT level. Data were expressed as HR and 95% confidence interval (CI). *P* < 0.05 was considered statistically significant. All statistical analyses were performed using JMP[®] 16.1.0 software (SAS Institute Inc., Cary, NC, USA).

Results

Participant characteristics

We enrolled 973 repeat visitors (487 females, 486 males) who underwent medical checkups during periods 1, 2, and 3 (Figure 1). Table 1 shows the participants' clinical characteristics. In period 1, the mean age was 52.5 years, mean BMI was 22.8 kg/m², and mean WC was 82.6 cm. In this period, 261 (27%) participants overall were diagnosed with MAFLD and 240 (25%) participants overall were diagnosed with NAFLD.

Table 2 shows the lifestyle habits of all participants during the 3 time periods. No significant changes in lifestyle habits were found before the COVID-19 pandemic (between periods 1 and 2) or during the pandemic (between periods 2 and 3).

Lifestyle habits related to MAFLD development

To determine the lifestyle characteristics of patients with newly developed or resolved MAFLD, we separately analyzed potential lifestyle habits associated with MAFLD development before and during the COVID-19 pandemic (Table 3). Before the pandemic, 22 patients developed new MAFLD from period 1 to period 2. Only the proportion of participants who consumed late-night meals was significantly higher among patients with newly developed MAFLD than among those without MAFLD (Table 3). During the pandemic, 44 patients developed new MAFLD from period 2 to period 3. Daily alcohol intake was significantly higher in patients with newly developed MAFLD than in those without MAFLD (mean 19.5 vs. 11.3 g/day, P = 0.001). Additionally, the proportions of current smokers and participants who ate 2 meals per day were significantly higher among patients who ate 3 times per day were significantly lower

among patients with newly developed MAFLD than among those without MAFLD (Table 3). Regarding MAFLD resolution, there were no significant differences in lifestyle habits between patients with or without MAFLD resolution in both the time before the COVID-19 pandemic and the time during the COVID-19 pandemic (data not shown).

Relationships between clinical and laboratory data and MAFLD development

We evaluated associations between clinical and laboratory data and MAFLD development in the time before the COVID-19 pandemic and the time during the pandemic (Appendix Table 1). Both before and during the pandemic, patients who developed MAFLD had significantly higher levels of lipid parameters than those without MAFLD: TG before, 123 vs. 78.0 mg/dL (P < 0.001); TG during, 115 vs. 76.4 mg/dL (P < 0.001); LDL-C before, 129 vs. 110 mg/dL (P = 0.003); and LDL-C during, 123 vs. 110 mg/dL (P = 0.002). Patients with newly developed MAFLD also had significantly higher values for several other parameters both before and during the pandemic: ALT before, 23.2 vs. 17.5 U/L (P = 0.003); ALT during, 24.4 vs. 17.5 U/L (P = 0.003); fasting plasma glucose before, 105 vs. 99.3 mg/d (P = 0.001); fasting blood glucose during, 102 vs. 99.0 mg/dL (P = 0.006); BMI before, 24.0 vs. 21.5 kg/m² (P < 0.001); BMI during, 23.9 vs. 21.5 kg/m² (P < 0.001); body weight before, 66.6 vs. 57.6 kg (P < 0.001); and WC during, 86.5 vs. 79.6 cm (P < 0.001) (Appendix Table 1).

Lifestyle habits as predictors associated with MAFLD development

Univariate and multivariate analyses were performed to explore lifestyle predictors of MAFLD development (Appendix Table 2 and Table 4). Univariate analyses showed that partaking in a routine late-night meal was a lifestyle predictor of MAFLD development in the time before the

COVID-19 pandemic (HR 2.47, 95% CI 1.05–5.82, P = 0.038). Lifestyle predictors of MAFLD development during the pandemic were higher daily alcohol intake (HR 1.02, 95% CI 1.01–1.04, P = 0.004), never smoking (HR 0.48, 95% CI 0.26–0.90, P = 0.021), current smoking (HR 2.34, 95% CI 1.08–5.08, P = 0.031), consuming 2 meals per day (HR 2.26, 95% CI 1.10–4.64, P = 0.026), and consuming 3 meals per day (HR 0.41, 95% CI 0.20–0.82, P = 0.012) (Appendix Table 2).

Multivariate adjusted HRs are presented in Table 4. In the time before the COVID-19 pandemic, partaking in a late-night meal remained an independent lifestyle predictor of MAFLD development (HR 2.54, 95% CI 1.02–6.36, P = 0.046), even after making adjustments for age, sex, BMI, Δ BMI, and ALT. During the pandemic, higher daily alcohol intake was the only independent lifestyle predictor of MALD development (HR 1.03, 95% CI 1.01–1.05, P = 0.008) after adjusting for the same factors (Table 4).

Lifestyle habits as predictors of MAFLD development considering interactive effects

We constructed a logistic regression model considering interactive effects between lifestyle factors (Appendix Table 3). In multivariate analysis, consuming a late-night meal was the only identified independent predictor of MAFLD development (HR 2.54, 95% CI 1.02–6.36, P = 0.046). There were no significant interactions between each factor and their respective differences.

Subgroup analysis according to age

Subgroup analysis of participants aged ≥ 60 years revealed no significant differences in lifestyle habits between patients with MALD development or without MALD development both before the pandemic and during the pandemic (Appendix Table 4). In participants < 60 years of age, there were also no significant differences in lifestyle habits between patients with or without MALD

development in the time before the COVID-19 pandemic. However, daily alcohol intake (mean 21.4 vs. 10.8 g/day, P < 0.001) and the proportion of participants who ate 2 times per day (28% vs 14%, P = 0.029) were significantly higher, and the proportion of participants who ate 3 times per day (69% vs 85.4%, P = 0.016) was significantly lower among participants who developed MAFLD during the COVID-19 pandemic in this younger age group (Table 5).

We also performed subgroup analysis of clinical factors in participants aged < 60 years (Appendix Table 5). Both before and during the pandemic, participants with newly developed MAFLD had significantly higher lipid parameters, ALT, fasting plasma glucose, BMI, body weight, and WC than participants without MAFLD. Moreover, during the COVID-19 pandemic, patients who developed new MAFLD had a significantly higher rate of hypertension, as well as higher age, diastolic BP, uric acid, AST, and GGT than those who did not develop MAFLD (Appendix Table 5).

Discussion

To the best of our knowledge, there have been no previous reports of increased MAFLD incidence associated with the lifestyle changes necessitated by the COVID-19 pandemic. In the current longitudinal study, we found that routine late-night meals before the COVID-19 pandemic and higher daily alcohol intake during the COVID-19 pandemic were independent lifestyle predictors of MAFLD development in a Japanese population.

Role of dietary habits in MAFLD development before and during the COVID-19 pandemic Habitual late-night meals were identified as an independent risk factor for NAFLD in a previous Japanese questionnaire-based study.²⁰ Nishi et al. reported that participants who ate before bedtime had a higher risk of NAFLD than those who did not (adjusted odds ratio [OR] 2.15, 95% CI 1.03–4.46).²⁰ High energy intake at night might decrease insulin sensitivity,²¹ which is strongly associated with fatty liver development.²² Additionally, recent evidence revealed that lifestyle changes post-COVID-19 were associated not only with obesity^{3, 4} and obesity-related diseases (NAFLD²³) but also with less healthy dietary habits.¹²

In this study, we confirmed routine late-night meals as a predictor of MAFLD development using multivariate analysis of interaction effects (Appendix Table 3). These results suggested that consuming late-night meals was the most important predictor of MAFLD development, regardless of the COVID-19 pandemic. Higher alcohol intake was identified as a new predictor of MAFLD development during the COVID-19 pandemic. A possible explanation is that participants with a predisposition towards late-night eating likely developed MAFLD by the second year of the study (2019), and increased alcohol intake during the COVID-19 pandemic may have masked the effects of late-night meals on MAFLD development during this later time period.

Role of alcohol intake in MAFLD development before and during the COVID-19 pandemic We identified higher daily alcohol intake as an independent lifestyle predictor of MAFLD development during the COVID-19 pandemic mainly among participants aged < 60 years, which represents a major proportion of the working-age population. One report showed that medical students in Portugal reduced their alcohol intake during the pandemic lockdown.²⁴ Conversely, a questionnaire-based survey of approximately 10,000 Japanese adults revealed that during the stay-at-home request, the number of people reporting increased alcohol intake was higher in participants with extended at-home time than in those without.²⁵

Because Alpers et al., who studied alcohol intake in the early stages of the COVID-19 pandemic, reported that people aged 30–39 years had a higher OR of increased drinking during lockdown than their oldest age group (OR 3.1, 95% CI 2.4–3.8),²⁷ we hypothesized that the impact of the COVID-19 pandemic on alcohol intake, as well as other lifestyle habits, would differ by

age.^{11, 12} As expected, our results were consistent with those of Alpers et al.²⁶ One possible reason for increased alcohol consumption is the increase in teleworking from home. A report from Norway revealed that working from home for > 15 h per week was significantly associated with increased alcohol consumption (OR 1.67, 95% CI 1.30–2.16).²⁷ Therefore, these data suggest that the increase in teleworking during the COVID-19 pandemic led to an increase in alcohol consumption among the working-age population.

Other factors affecting MAFLD development before and during the COVID-19 pandemic

Current smoking and consuming 2 meals per day were identified as predictors of MAFLD development during the COVID-19 pandemic only on univariate analysis (Appendix Table 2). A recent 31-year cohort study demonstrated that individuals with persistent exposure to passive smoking between childhood and adulthood had the highest risk of fatty liver (relative risk 1.99, 95% CI 1.14–3.45).²⁸ Moreover, Alhussain et al. found that reduced meal frequency associated with irregular meal timing could cause weight gain and increase the levels of hunger-related hormones (glucagon-like peptide 1 and peptide YY), ultimately leading to metabolic disturbance.²⁹ Future large-scale prospective studies are necessary to clarify whether these lifestyle habits are indeed involved in MAFLD development.

Study strengths and weaknesses

The main strength of our study lies in its research design. We designed a longitudinal study to provide objective evidence regarding the effects of COVID-19–related lifestyle changes over time in the same individuals. Nevertheless, several limitations must be considered when interpreting our results. First, this was a retrospective, single-center study performed in Osaka, second largest city in Japan. Interestingly, recent study demonstrated that there are significant differences in the

proportion of Japanese NAFLD patients by region, but not by population-setting (urban or rural) or by per-person income data.³⁰ Second, selection bias was a major limitation. Our health checkup facility was temporarily closed from April 8, 2020 to May 31, 2020 to prevent the spread of COVID-19. Consequently, 30% of patients examined consecutively in 2018 and 2019 were not examined in 2020. This resulted in a smaller number of individuals who visited in all 3 periods and were thus included in this study. Third, we did not use the International Physical Activity Questionnaire, an internationally standardized questionnaire.³¹ Thus, the assessment of physical activity levels may be insufficient to draw reliable conclusions about this potential predictive factor. Finally, most participants were healthy enough to engage in work and were sufficiently conscious of their health to voluntarily undergo health checkups.³² Therefore, the study results may not apply to individuals who are less healthy.

In conclusion, we demonstrated that routinely partaking in late-night meals during the time before the COVID-19 pandemic and higher daily alcohol intake during the COVID-19 pandemic were independent lifestyle predictors for MAFLD development. These results highlight the need to avoid excessive alcohol intake as the pandemic continues to affect lifestyle patterns.

This article is protected by copyright. All rights reserved

- The Ministry of Health, Labour, and Welfare. Coronavirus disease (COVID-19) situation report in Japan. Available online:
 - https://wwwmhlwgojp/stf/covid-19/kokunainohasseijoukyouhtml (accessed on 31 March 2021).
- Azuma K, Nojiri T, Kawashima M, et al. Possible favorable lifestyle changes owing to the coronavirus disease 2019 (COVID-19) pandemic among middle-aged Japanese women: An ancillary survey of the TRF-Japan study using the original "Taberhythm" smartphone app. PLoS One 2021;16:e0248935.
- Coulthard H, Sharps M, Cunliffe L, et al. Eating in the lockdown during the Covid 19 pandemic; self-reported changes in eating behaviour, and associations with BMI, eating style, coping and health anxiety. Appetite 2021;161:105082.
- 4. Mason TB, Barrington-Trimis J, Leventhal AM. Eating to cope with the COVID-19 pandemic and body weight change in young adults. J Adolesc Health 2021;68:277-283.
- Younossi Z, Tacke F, Arrese M, et al. Global perspectives on nonalcoholic fatty liver disease and nonalcoholic steatohepatitis. Hepatology 2019;69:2672-2682.
- 6. Chalasani N, Younossi Z, Lavine JE, et al. The diagnosis and management of nonalcoholic fatty liver disease: Practice guidance from the American Association for the Study of Liver Diseases. Hepatology 2018;67:328-357.
- European Association for the Study of the Liver (EASL); European Association for the Study of Diabetes (EASD); European Association for the Study of Obesity (EASO).
 EASL-EASD-EASO Clinical Practice Guidelines for the management of non-alcoholic fatty liver disease. J Hepatol 2016;64:1388-1402.

- Eslam M, Newsome PN, Sarin SK, et al. A new definition for metabolic dysfunction-associated fatty liver disease: An international expert consensus statement. J Hepatol 2020;73:202-209.
 - Shiha G, Korenjak M, Eskridge W, et al. Redefining fatty liver disease: an international patient perspective. Lancet Gastroenterol Hepatol 2021;6:73-79.
 - Yoshida S, Kurajoh M, Fukumoto S, et al. Association of plasma xanthine oxidoreductase activity with blood pressure affected by oxidative stress level: MedCity21 health examination registry. Sci Rep 2020;10:4437.
- 11. Tanaka N, Hamamoto Y, Kurotobi Y, et al. Lifestyle changes as a result of COVID-19 containment measures: Bodyweight and glycemic control in patients with diabetes in the Japanese declaration of a state of emergency. J Diabetes Investig 2021;12:1718-1722.
- Shimpo M, Akamatsu R, Kojima Y, et al. Factors Associated with Dietary Change since the Outbreak of COVID-19 in Japan. Nutrients 2021;13:2039.
- American Diabetes Association. Diagnosis and classification of diabetes mellitus. Diabetes Care 2010;33:S62-69.
- 14. Teramoto T, Sasaki J, Ueshima H, et al. Diagnostic criteria for dyslipidemia. Executive summary of Japan Atherosclerosis Society (JAS) guideline for diagnosis and prevention of atherosclerotic cardiovascular diseases for Japanese. J Atheroscler Thromb 2007;14:155-158.
- 15. Umemura S, Arima H, Arima S, et al. The Japanese Society of Hypertension Guidelines for the Management of Hypertension (JSH 2019). Hypertens Res 2019;42:1235-1481.
- Sterling RK, Lissen E, Clumeck N, et al. Development of a simple noninvasive index to predict significant fibrosis in patients with HIV/HCV coinfection. Hepatology 2006;43:1317-1325.

- Hamaguchi M, Kojima T, Itoh Y, et al. The severity of ultrasonographic findings in nonalcoholic fatty liver disease reflects the metabolic syndrome and visceral fat accumulation. Am J Gastroenterol 2007;102:2708-2715.
- Fujii H, Nishimoto N, Miyano M, et al. The alcohol use disorders. Identification Test (AUDIT) score is useful for predicting alcohol consumption. Nihon Arukoru Yakubutsu Igakkai Zasshi 2016;51:293-301.
- Fujii H, Fukumoto S, Enomoto M, et al. The FibroScan-aspartate aminotransferase score can stratify the disease severity in a Japanese cohort with fatty liver diseases. Sci Rep 2021;11:13844.
- 20. Nishi T, Babazono A, Maeda T, et al. Effects of eating fast and eating before bedtime on the development of nonalcoholic fatty liver disease. Popul Health Manag 2016;19:279-283.
- Morgan LM, Shi JW, Hampton SM, et al. Effect of meal timing and glycaemic index on glucose control and insulin secretion in healthy volunteers. Br J Nutr 2012;108:1286-1291.
- 22. Gastaldelli A, Kozakova M, Højlund K, et al. Fatty liver is associated with insulin resistance, risk of coronary heart disease, and early atherosclerosis in a large European population. Hepatology 2009;49:1537-1544.
- 23. Shanmugam H, Di Ciaula A, Di Palo DM, et al. Multiplying effects of COVID-19lockdown on metabolic risk and fatty liver. Eur J Clin Invest 2021;51:e13597.
- Vasconcelos M, Crego A, Rodrigues R, et al. Effects of the COVID-19 mitigation measures on alcohol consumption and binge drinking in college students: A longitudinal survey. Int J Environ Res Public Health. 2021;18:9822.
- 25. Nishijima C, Miyagawa N, Tsuboyama-Kasaoka N, et al. Association between lifestyle changes and at-home hours during and after the state of emergency due to the COVID-19 pandemic in Japan. Nutrients. 2021;13:2698.

- 26. Alpers SE, Skogen JC, Maeland S, et al. Alcohol consumption during a pandemic lockdown period and change in alcohol consumption related to worries and pandemic measures. Int J Environ Res Public Health. 2021;18:1220.
 - Nielsen MB, Christensen JO, Knardahl S. Working at home and alcohol use. Addict Behav Rep. 2021;14:100377.
 - Wu F, Pahkala K, Juonala M, et al. Childhood and adulthood passive smoking and nonalcoholic fatty liver in midlife: A 31-year cohort study. Am J Gastroenterol. 2021;116:1256-1263.
 - 29. Alhussain MH, Macdonald IA, Taylor MA. Irregular meal-pattern effects on energy expenditure, metabolism, and appetite regulation: a randomized controlled trial in healthy normal-weight women. Am J Clin Nutr. 2016;104:21-32.
 - Ito T, Ishigami M, Zou B, et al. The epidemiology of NAFLD and lean NAFLD in Japan: a meta-analysis with individual and forecasting analysis, 1995-2040. Hepatol Int. 2021 Apr;15(2):366-379. doi: 10.1007/s12072-021-10143-4. Epub 2021 Feb 12.
 - Craig CL, Marshall AL, Sjöström M, et al. International physical activity questionnaire:
 12-country reliability and validity. Med Sci Sports Exerc 2003;35:1381-1395.
 - 32. Moriya A, Iwasaki Y, Ohguchi S, et al. Roles of alcohol consumption in fatty liver: a longitudinal study. J Hepatol 2015;62:921-927.

This article is



Figure legends

Figure 1. Study flow chart. COVID-19: coronavirus disease 2019.

	Variables	Period 1	Period 2	Period 3
	Age (y)	52.5 (10.4)	53.6 (10.4)	54.6 (10.4)
	Sex $(female/male)^{+}$	487/486	487/486	487/486
	BMI (kg/m^2)	22.8 (3.4)	22.8 (3.5)	22.9 (3.5)
	WC (cm)	82.6 (9.5)	83.3 (9.7)	83.6 (9.9)
	MAFLD	261 (27)	252 (26)	272 (28)
_	NAFLD [®]	240 (25)	229 (24)	246 (25)
	Hypertension [®]	206 (21)	241 (25)	317 (33)
	DM ³	75 (8)	85 (9)	97 (10)
	Dyslipidemia [®] ,	441 (45)	427 (44)	510 (52)
	SBP (mmHg)	118.3 (15.5)	119.8 (15.4)	121 (17.9)
	DBP (mmHg)	74.2 (10.7)	74.8 (10.6)	75.6 (12.0)
	TC $(mg/dL)_{\star}^{\dagger}$	198.8 (31.9)	197.9 (32.1)	204 (33.9)
	TG (mg/dL)	97.6 (77.6)	97.5 (74.8)	99.3 (66.8)
	$HDL-C (mg/dL)_{*}$	61.3 (15.9)	61.0 (15.8)	61.5 (16.1)
	LDL-C (mg/dL)	112.3 (28.0)	112.9 (27.9)	116.0 (29.0)
	Non-HDL-C (mg/dL)	137.5 (30.9)	136.8 (31.4)	142.8 (32.1)
	Uric acid (mg/dL)	5.3 (1.4)	5.3 (1.3)	5.3 (1.3)
	$AST(U/L)_{\star}$	21.5 (7.6)	21.9 (8.3)	22.1 (7.9)
	ALT $(U/L)_{*}$	21.2 (13.5)	21.7 (14.0)	22.2 (14.3)
	GGT (U/L) *	34.4 (41.2)	34.2 (36.8)	34.2 (37.7)
	Albumin $(g/dL)'$	4.2 (0.3)	4.3 (0.3)	4.3 (0.3)
	Platelet count $(\times 10^{\circ}/L)$	229 (54.8)	230 (57.7)	239 (60.7)
	Fasting plasma glucose (mg/dL)	103 (14.4)	103 (15.6)	103 (15.2)
	HbA1c (%)	5.7 (0.5)	5.7 (0.5)	5.7 (0.5)
	FIB-4 index score	1.21 (0.57)	1.24 (0.60)	1.22 (0.59)

Table 1. Clinical characteristics of participants (N=973)

ALT: alanine aminotransferase; AST: aspartate aminotransferase; BMI: body mass index; DM: diabetes mellitus; DBP: diastolic blood pressure; FIB-4: Fibrosis-4; GGT: gamma-glutamyl transferase; HDL-C: high-density lipoprotein cholesterol; LDL-C: low-density lipoprotein cholesterol; MAFLD: metabolic dysfunction-associated fatty liver disease; NAFLD: non-alcoholic fatty liver disease; SBP: systolic blood pressure; TC: total cholesterol; TG: triglycerides; WC: waist circumference.

[†] Mean (standard deviation). [‡] Number. § Number (%).

Variables	Period 1	Period 2	Period 3	<i>P</i> value Period 1 vs. Period 2	<i>P</i> value Period 2 vs. Period 3
Alcohol intake (g/day)	13.0 (17.7)	12.8 (17.2)	11.7 (17.0)	0.99	0.150
Smoking status *					
Never	572 (59)	572 (59)	567 (58)	0.99	1.00
Past	282 (29)	286 (29)	302 (31)		
Current	119 (12)	115 (12)	104 (11)		
Exercise (times/week) ⁺					
< 1	547 (56)	510 (52)	495 (51)	0.21	0.30
1–2	275 (28)	302 (31)	278 (29)		
\geq 3	<u>1</u> 51 (16)	161 (17)	200 (20)		
Sleep duration (h/night) [‡]				
< 6	360 (37)	356 (37)	328 (33)	0.99	0.28
6–8	583 (60)	597 (61)	621 (64)		
≥ 8 *	30 (3)	20 (2)	24 (3)		
Meals per day $\frac{1}{2}$					
1	4 (0.4)	5 (0.5)	5 (0.5)	0.38	0.60
2	118 (12)	135 (14)	122 (12.5)		
3 *	851 (87.6)	833 (85.5)	846 (87.0)		
Late-night meal [*]					
Yes	266 (27)	254 (26)	219 (23)	0.76	0.116
No	707 (73)	707 (74)	754 (77)		

Table 2. Lifestyle habits of participants (N=973)

[†] Mean (standard deviation). ‡ Number (%).

2

	Before COVID-19 pandemic (2018–2019)			During COVID-19 pandemic (2019–2020)		
Variables [†]	No (n=690)	Yes (n=22)	P value	No (n=677)	Yes (n=44)	P value
Alcohol intake (g/day) [†]	12.0 (16.7)	15.8 (20.2)	0.30	11.3 (16.1)	19.5 (19.1)	0.001
Smoking status [‡]						
Never	433 (63)	14 (64)	1.00	428 (63)	20 (45)	0.024
Past	180 (26)	7 (32)	0.62	182 (27)	15 (34)	0.30
Current	77 (11)	1 (4)	0.50	67 (10)	9 (21)	0.039
Exercise $(times/week)^{\ddagger}$						
< 1	376 (54)	15 (68)	0.28	345 (51)	24 (55)	0.76
1–2	203 (29)	4 (18)	0.32	212 (31)	16 (36)	0.51
≥3	111 (17)	3 (14)	1.00	120 (18)	4 (9)	0.21
Sleep duration $(h/night)^{\ddagger}$						
< 6	259 (38)	7 (31)	0.66	254 (38)	15 (34)	0.75
6–8	407 (59)	14 (64)	0.83	406 (60)	28 (64)	0.75
≥ 8	24 (3)	1 (5)	0.55	17 (2)	1 (2)	1.00
Meals per day [‡]						
1	2 (0.3)	0 (0)	1.00	3 (0.4)	1 (2)	0.22
2	84 (12)	1(5)	0.50	87 (13)	11 (25)	0.037
3	604 (87.7)	21 (95)	0.50	587 (86.6)	32 (73)	0.022
Late-night meal [‡]						
Yes	174 (25)	10 (45)	0.046	177 (26)	10 (23)	0.86

Table 3. Relationship between patient lifestyle habits and MAFLD development

COVID-19, coronavirus disease 2019; MAFLD: metabolic dysfunction–associated fatty liver disease. [†] Mean (standard deviation). [‡] Number (%).

Note: Significant results are highlighted in bold.

	Before COVID-19 p (2018–2019)	andemic	During COVID-19 pandemic (2019–2020)		
Variables	Hazard ratio (95% CI)	P value	Hazard ratio (95% CI)	P value	
Age (y)	0.99 (0.94–1.03)	0.56	0.98 (0.95-1.02)	0.28	
Sex (male)	1.08 (0.35-3.32)	0.90	1.17 (0.52–2.64)	0.71	
BMI (kg/m^2)	1.35 (1.15–1.58)	<0.001	1.38 (1.22–1.57)	<0.001	
$\Delta BMI (kg/m^2)$	2.07 (1.13-3.78)	0.013	2.94 (1.84-4.71)	<0.001	
ALT (U/L)	1.05 (1.01-1.10)	0.025	1.03 (1.01–1.06)	0.013	
LIFESTYLE HABITS					
Alcohol intake (g/day)) 1.01 (0.98–1.04)	0.38	1.03 (1.01–1.05)	0.008	
Smoking status					
Never	4.59 (0.48-43.6)	0.185	0.77 (0.28–2.14)	0.61	
Past	3.46 (0.37-32.7)	0.28	0.99 (0.36–2.71)	0.98	
Current	N.A.		N.A.		
Meals per day					
1	N.A.	0.99	4.29 (0.25-73.5)	0.31	
2	0.35 (0.04-2.93)	0.33	2.23 (0.95-5.22)	0.065	
3	N.A.		N.A.		
Late-night meal †					
Yes	2.54 (1.02-6.36)	0.046	0.46 (0.20–1.07)	0.070	

Table 4. Results of multivariate analyses: Predictors of MAFLD development

ALT: alanine aminotransferase; BMI: body mass index; CI: confidence interval; COVID-19: coronavirus disease 2019; MAFLD: metabolic dysfunction–associated fatty liver disease; N.A.: not applicable.

Dinner within 2 hours before bedtime at least 3 times per week.

Note: Significant results are highlighted in bold.

Before COV (201		OVID-19 pai 2018–2019)	ndemic	During COVID-19 pand (2019–2020)		lemic
$\mathbf{Variables}^{\dagger}$	No (n=506)	Yes (n=18)	P value	No (n=473)	Yes (n=36)	P value
Alcohol intake (g/day) [‡]	11.6 (16.4)	14.1 (20.2)	0.52	10.8 (15.7)	21.4 (19.8)	<0.001
Smoking status						
Never	324 (64)	13 (72)	0.48	305 (64)	18 (50)	0.105
Past	118 (23)	4 (22)	0.91	115 (24)	10 (28)	0.69
Current	64 (13)	1 (6)	0.77	53 (12)	8 (22)	0.061
Exercise (times/week)						
1	314 (62)	14 (79)	0.22	271 (57)	22 (61)	0.73
1–2	138 (27)	3 (17)	0.42	149 (32)	11 (31)	1.00
\geq 3	54 (11)	1 (4)	0.71	53 (11)	3 (8)	0.79
Sleep duration (h/night)						
< 6	204 (40)	7 (39)	1.00	189 (40)	13 (36)	0.73
6–8	290 (57)	11 (61)	0.81	278 (59)	22 (61)	0.86
≥ 8	12 (3)	0 (0)	1.00	6(1)	1 (3)	0.40
Meals per day						
1	2(1)	0 (0)	1.00	3 (0.6)	1 (3)	0.26
2	70 (13)	1 (6)	0.49	65 (14)	10 (28)	0.029
3	434 (86)	17 (94)	0.49	405 (85.4)	25 (69)	0.016
Late-night meal						
Yes	130 (26)	8 (44)	0.099	148 (31)	9 (25)	0.58

Table 5. Relationship between patient lifestyle habits and MAFLD development in
participants aged < 60 years (N=707)</th>

COVID-19: coronavirus disease 2019; MAFLD: metabolic dysfunction-associated fatty liver disease.

[†]Number (%). [‡]Mean (standard deviation).

Note: Significant results are highlighted in bold.